

ZBW

A Service of

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Niimi, Yoko

## **Working Paper**

# Does providing informal elderly care hasten retirement? Evidence from Japan

ADBI Working Paper, No. 730

### **Provided in Cooperation with:**

Asian Development Bank Institute (ADBI), Tokyo

Suggested Citation: Niimi, Yoko (2017): Does providing informal elderly care hasten retirement? Evidence from Japan, ADBI Working Paper, No. 730, Asian Development Bank Institute (ADBI), Tokyo

This Version is available at: https://hdl.handle.net/10419/163224

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



https://creativecommons.org/licenses/by-nc-nd/3.0/igo/





# **ADBI Working Paper Series**

# DOES PROVIDING INFORMAL ELDERLY CARE HASTEN RETIREMENT? EVIDENCE FROM JAPAN

Yoko Niimi

No. 730 April 2017

**Asian Development Bank Institute** 

Yoko Niimi is research associate professor at the Asian Growth Research Institute in Fukuoka, Japan.

The views expressed in this paper are the views of the author and do not necessarily reflect the views or policies of ADBI, ADB, its Board of Directors, or the governments they represent. ADBI does not guarantee the accuracy of the data included in this paper and accepts no responsibility for any consequences of their use. Terminology used may not necessarily be consistent with ADB official terms.

Working papers are subject to formal revision and correction before they are finalized and considered published.

The Working Paper series is a continuation of the formerly named Discussion Paper series; the numbering of the papers continued without interruption or change. ADBI's working papers reflect initial ideas on a topic and are posted online for discussion. ADBI encourages readers to post their comments on the main page for each working paper (given in the citation below). Some working papers may develop into other forms of publication.

ADB recognizes "Korea" as the Republic of Korea.

Suggested citation:

Niimi, Y. 2017. Does Providing Informal Elderly Care Hasten Retirement? Evidence from Japan. ADBI Working Paper 730. Tokyo: Asian Development Bank Institute. Available: https://www.adb.org/publications/does-providing-informal-elderly-care-hasten-retirement-evidence-japan

Please contact the authors for information about this paper.

Email: niimi@agi.or.jp

The empirical work undertaken in this paper utilizes microdata from the Preference Parameters Study of Osaka University's 21st Century COE Program "Behavioral Macrodynamics Based on Surveys and Experiments" and its Global COE Project "Human Behavior and Socioeconomic Dynamics." I acknowledge the program/project's contributors—Yoshiro Tsutsui, Fumio Ohtake, and Shinsuke Ikeda. I am also grateful to Axel Boersch-Supan, Ngee-Choon Chia, Erbiao Dai, Shoshana Grossbard, Charles Yuji Horioka, Minchung Hsu, Kenichi Imai, Yoshihiko Kadoya, John Laitner, Astghik Mavisakalyan, Andy McKay, Peter Morgan, Edward Norton, Eric D. Ramstetter, Hiroshi Sakamoto, Kazuki Tamura, Wataru Suzuki, Tien Manh Vu, Yanfei Zhou, and other participants of the ADBI-AGI Workshop on Aging in Asia, the Second HiHER (Hiroshima Institute of Health Economics Research) Seminar, and the AGI (Asian Growth Research Institute) Staff Seminar for their valuable comments. This work was supported by JSPS (Japan Society for the Promotion of Science) KAKENHI Grant Number 15H01950, a project grant from the Asian Growth Research Institute, and a grant from the MEXT Joint Usage/Research Center at the Institute of Social and Economic Research, Osaka University.

Asian Development Bank Institute Kasumigaseki Building, 8th Floor 3-2-5 Kasumigaseki, Chiyoda-ku Tokyo 100-6008, Japan

Tel: +81-3-3593-5500 Fax: +81-3-3593-5571 URL: www.adbi.org E-mail: info@adbi.org

#### Abstract

This paper examines the implications of providing care to elderly parents for adult children's retirement plans using microdata from a Japanese survey. We find no significant effect of caregiving on family caregivers' planned retirement age if we do not take into account caregiving intensity but find a negative and significant effect on retirement plans for intensive caregivers, particularly among women. These findings suggest that relying on family members to provide elderly care can pose a serious challenge to the ongoing efforts of the government to promote the labor supply of women and the elderly to address the shrinkage of the working-age population in Japan. The estimation results suggest that ensuring access to formal care services can help family members reconcile their paid work with caregiving requirements, thereby alleviating the adverse effect of caregiving on their retirement plans. The results also suggest that the financial burden of formal care services could require caregivers to postpone retirement in some cases.

**Keywords:** aging, caregiving, elderly care, informal care, Japan, labor supply, long-term care, parental care, retirement

**JEL Classification:** D10, J14, J26

# **Contents**

1.	INTR	ODUCTION	1
2.	CON	CEPTUAL FRAMEWORK	2
3.	LITE	RATURE REVIEW	4
4.	DATA	A AND ESTIMATION STRATEGY	6
	4.1 4.2 4.3	Data Estimation Strategy Empirical Specification	7
5.	EMPI	RICAL RESULTS	12
	5.1 5.2 5.3	Descriptive Statistics Endogeneity of Caregiving and Formal Care Usage Variables Main Results	16
6.	CON	CLUSIONS	24
REFE	ERENC	ES	26
APPE	ENDIX:	IV ESTIMATION RESULTS	29

# 1. INTRODUCTION

Japan has experienced an unprecedented speed of population aging over the past few decades. The share of the population aged 65 and above in Japan (9.9%) was the lowest among the then member countries of the Organisation for Economic Co-operation and Development (OECD) until as recently as 1984, but it had become the highest (20.2%) by 2005. It is estimated to be 27.5% in 2016 and expected to reach 30% by 2024 (National Institute of Population and Social Security Research 2012). Moreover, the aged dependency ratio (the ratio of the elderly population to the working-age population) increased even more rapidly from 14.6% in 1984 to 42.4% in 2014 in Japan. Improvements in longevity as well as a significant decline in the fertility rate over the years contributed to this rapid population aging in Japan.

Combined with changes in family structure with a downward trend in the parent-child co-residence rate,<sup>3</sup> these demographic trends are likely to reduce the availability of family members to provide elderly care and impose a greater burden on a smaller number of family caregivers per elderly person.<sup>4</sup> This poses significant challenges to Japan where elderly care has traditionally taken place within the family setting. While Japan introduced a mandatory long-term care insurance (LTCI) program in 2000 to promote the greater independence of the elderly in their daily lives and to reduce the burden of elderly care on family members, some studies show that adult children continue to be the most common source of elderly care in Japan (e.g., Hanaoka and Norton 2008; Long, Campbell, and Nishimura 2009).

One of the important costs of the increasing demand for elderly care is a possible reduction in the labor supply of family caregivers. Caregiving can, in principle, affect caregivers' labor market behavior at the extensive or intensive margins. Changes at the extensive margin include quitting work temporarily or retiring early while changes at the intensive margin include adjusting work hours (for example by switching from a full-time to a part-time job), taking on fewer responsibilities, and/or forgoing a promotion (Van Houtven, Coe, and Skira 2013). This paper examines adjustments at the extensive margin and pays particular attention to the effect of providing care to elderly parents on the retirement plans of adult children. Given that the demand for parental care provision tends to increase with age, it is possible that the need for adult children to provide care to their elderly parents is concentrated around the period when retirement is a possible option for labor market exit (Meng 2012).

Taking early retirement for caregiving reasons can cause serious financial costs for caregivers as it is likely to affect their lifetime income not only by making them forgo the income they could have earned until the mandatory retirement age but also by reducing pension entitlements as well as the lump sum severance payments that are commonly paid upon retirement in the case of Japan. As a result, earlier retirement due to caregiving commitments is likely to have long-term repercussions for the livelihood of family caregivers. Such financial consequences are likely to be greater for women than for men as women are more likely to be the bearer of the burden of elderly care and

OECD data (https://data.oecd.org/pop/elderly-population.htm, accessed on 24 September 2015).

<sup>&</sup>lt;sup>2</sup> The data on aged dependency ratios are from the Statistics Bureau, the Ministry of Internal Affairs and Communications (http://www.stat.go.jp/data/jinsui/2.htm#annual, accessed on 29 July 2016).

The share of elderly persons who live with their children (married and unmarried) decreased by more than 40% from 69.0% to 40.6% between 1980 and 2014 in Japan (Niimi 2016).

<sup>&</sup>lt;sup>4</sup> Throughout this paper, "caregivers" refer to those who provide informal elderly care, not to care workers who provide professional care.

they tend to work fewer years than men as women tend to have more fragmented employment histories due to their other family responsibilities.

If elderly care provision increases the likelihood of family caregivers taking early retirement, this could also cause serious conflicts with the government's policy of promoting the labor market participation of women as well as elderly persons in response to the shrinkage of the working-age population in Japan. It is therefore critical for policymakers to gauge the effect of elderly care provision on family caregivers' retirement decisions, and if caregiving does indeed have a negative effect, to formulate appropriate measures for preventing family members from having to take early retirement in order to accommodate caregiving requirements.

Japan is an interesting case to study as the increasing use of formal care services since the launch of the LTCI program in 2000 provides an opportunity to examine whether the provision of formal care services helps alleviate or eliminate any adverse effects of caregiving on family caregivers' retirement plans. Moreover, it would be interesting to see whether or not previous findings obtained for other advanced economies hold in the case of Japan where filial obligation remains relatively strong and caring for elderly parents has therefore traditionally taken place within the family setting. Given that many Asian countries that share a common culture with Japan are expected to experience a significant aging of their populations over the next few decades, analyzing the Japanese case can also shed light on important issues that Asian countries need to take into account when preparing themselves for population aging and the advent of an aging society.

The main aim of this paper is therefore to contribute to a better understanding of the impact of elderly care provision on family caregivers' retirement plans. This paper examines specifically the effect of providing care to elderly parents on the planned retirement age of adult children in Japan. Such an analysis will help policymakers become better informed about the potential impact of caregiving on family caregivers' retirement decisions as well as on their retirement security. While there has been a growing literature that analyzes the effect of providing elderly care on family caregivers' labor market outcomes, empirical evidence on the effect of caregiving on their retirement remains limited not only in Japan but also in other parts of the world. This paper attempts to fill this gap in the literature using microdata from the "Preference Parameters Study" of Osaka University, a nationally representative survey conducted in Japan.

The rest of the paper is organized as follows. Section 2 provides a conceptual framework for analyzing the effect of elderly care provision on the labor supply of family caregivers while Section 3 reviews the relevant literature. Section 4 describes the data and estimation strategy. Estimation results are presented in Section 5. Section 6 summarizes the main findings and discusses some policy implications.

# 2. CONCEPTUAL FRAMEWORK

There have been an increasing number of empirical studies that examine the employment effects of elderly care provision in recent decades. Empirical analyses are based mainly on the idea that the utility maximization decision in the presence of care needs is in principle very similar to the standard labor market participation decision where labor market participation is observed if, and only if, the offered wage exceeds the reservation wage (Heitmueller and Inglis 2007). It is thus hypothesized that the effect of caregiving on labor supply will be the net impact of two opposing forces, namely substitution and income effects (Carmichael and Charles 1998, 2003). With

time being scarce, caregiving responsibilities will tend to increase family caregivers' reservation wages and reduce their labor supply (substitution effect) while greater expenditures associated with elderly care may reduce their disposable incomes and induce them to remain in the labor market (income effect).

In addition to these two main effects, there are additional effects of caregiving on the labor supply of family caregivers, namely respite and discrimination effects (Carmichael and Charles 1998, 2003). The respite effect exists when family caregivers want to take a break from caregiving responsibilities through their engagement in employment, provided that the hours and commitments are compatible with caregiving demands. The respite effect is likely to reduce the reservation wage and counteract the substitution effect with regard to the decision on labor market participation, though possibly not with regard to the decision on the number of hours of work (Carmichael and Charles 1998). As for the discrimination effect, family caregivers may experience discrimination in terms of wages and/or promotion because of their higher flexibility requirements and lower reliability due to caregiving commitments (e.g., higher absence and sickness rates) and thus reduce their labor supply (Carmichael and Charles 1998; Heitmueller and Inglis 2007). Even without such discrimination, family caregivers themselves might prefer jobs with less demanding responsibilities and more flexible working arrangements, which enable them to combine work with caregiving responsibilities (Carmichael and Charles 1998).

In sum, the employment effect of caregiving will be positive when the income effect dominates the substitution effect and negative when the latter dominates the former. Which effect outweighs the other is theoretically ambiguous and is an empirical question. The employment effect of caregiving will also depend on the size of the respite (positive) and discrimination (negative) effects. Moreover, the overall direction of the effect will depend on the degree of caregiving intensity (Carmichael and Charles 1998, 2003), how strongly caregivers are attached to the labor market (Carmichael and Charles 2003), and the availability of formal care services (Kotsadam 2012; Michaud, Heitmueller, and Nazarov 2010).

In the case of Japan, the LTCI program introduced in 2000 has universal coverage and everyone aged 65 and above as well as those under 65 but with aging-related disabilities are entitled to receive necessary care services regardless of their income level or the availability of family caregivers as long as they are certified as requiring support or long-term care. The Japanese LTCI program does not provide cash allowances to the elderly to support their informal caregivers, but it covers the cost of services purchased from the formal sector once they are certified as requiring care. A professional care manager provides a personal care plan and care recipients can choose what services to receive and from which provider to receive these services subject to a 10% co-payment (Tsutsui and Muramatsu 2005).

Reflecting the Japanese context, the key hypotheses that we will test in this paper are as follows: (i) Adult children who provide care to their elderly parents and/or parents-in-law plan to retire at a younger age than those without such responsibilities; (ii) given that women are traditionally weakly attached to the labor market, the negative effect of caregiving on caregivers' retirement is greater for female caregivers than their male counterparts; (iii) the adverse employment effect of caregiving is greater for caregivers who play the main caregiving role than for those with less caregiving responsibility; (iv) given the availability of formal care services under the LTCI program, the income effect of caregiving is limited in the case of Japan; and (v) the usage of formal care services by care recipients helps alleviate or eliminate the negative effect of caregiving on their adult children's retirement.

# 3. LITERATURE REVIEW

There has been a growing literature that examines the employment effects of providing elderly care in recent decades. <sup>5</sup> The results from empirical studies have been somewhat mixed, but more consensus has been reached on the adverse effect of caregiving on labor supply (particularly in the case of women) for intensive caregivers or co-residential caregivers (e.g., Casado-Marín, García-Gómez, and López-Nicolás 2011; Ettner 1995; Heitmueller 2007; Kotsadam 2012; Lilly, Laporte, and Coyte 2010; Michaud, Heitmueller, and Nazarov 2010; Nguyen and Connelly 2014). Similar results have been obtained with respect to the effect of caregiving on wages (e.g., Carmichael and Charles 1998, 2003; Heitmueller and Inglis 2007). These findings suggest that caregivers are heterogeneous in both their caregiving inputs and associated labor market responses, thereby underscoring the importance of controlling for caregiving intensity when analyzing the employment effects of caregiving (Lilly, Laporte, and Coyte 2010).

Despite the growing literature on the employment effects of caregiving over the last few decades, there remain some gaps in the literature, particularly in terms of geographical coverage outside the United States (US) and European countries as well as empirical work on the effect of caregiving on family caregivers' retirement.

As far as the geographical coverage of the literature is concerned, most studies have so far been based largely on data on the US or European countries, and the number of studies that look at the issue outside these countries remains relatively limited (e.g., Berecki-Gisolf *et al.* (2008), Bittman, Hill, and Thomson (2007), Leigh (2010), and Nguyen and Connelly (2014) on Australia; Schneider *et al.* (2013) on Austria; Lilly, Laporte, and Coyte (2010) on Canada; and Do (2008) on the Republic of Korea). Japan is no exception and there is a paucity of empirical evidence on Japan. Sakai and Sato (2007) found, based on panel data analysis, that while the presence of family members in need of care has a negative effect on the probability of having a regular job or being self-employed for the male sample, it has a negative effect on the probability of having a nonregular job for the female sample. They also examined whether the introduction of the LTCI program alleviates the adverse effect of caregiving on employment, but their difference-in-difference analysis did not generate any conclusive results.

Shimizutani, Suzuki, and Noguchi (2008) similarly assessed the effect of the introduction of the LTCI program on female labor market participation. Their difference-in-difference estimates suggest that the introduction of the LTCI program had no effect in 2001 but had a large and positive effect in 2002 at both the extensive and intensive margins. In addition, Sugawara and Nakamura (2004) found that the LTCI program alleviates the negative effect of caregiving requirements (measured by the presence of an elderly person in need of care in the household) on the labor supply of women by comparing the estimated effects of caregiving using repeated cross-sectional data for 1998, 2004, and 2010.

Yamada and Shimizutani (2015) found an adverse effect of caregiving on the labor market outcomes of main caregivers at both the extensive and intensive margins by estimating an instrumental variable (IV) model using cross-sectional data. The negative effect of caregiving was found to be greater for female caregivers than for male caregivers. Their analysis also shows that the LTCI program helps mitigate the

\_

See Bauer and Spousa-Poza (2015) and Lilly, Laporte, and Coyte (2007) for a comprehensive survey of the literature on the impact of caregiving on caregivers' employment.

negative impact of caregiving on main caregivers' labor supply but only to a limited extent. On the other hand, Kan and Kajitani (2014) found that while the introduction of the LTCI program helps reduce the hours of caregiving among highly educated women, the reduction does not lead to an increase in their working hours, according to their difference-in-difference estimates. Finally, Oshio and Usui (2016) found that elderly care provision has little impact on female labor supply after controlling for the endogeneity of caregiving or individual unobserved time-invariant heterogeneity.

Another gap in the literature is the limited empirical work on the effect of caregiving on caregivers' retirement behavior. As reviewed above, existing studies on Japan hardly look at the effect of caregiving obligations on caregivers' retirement. One exception, which is somewhat related, is the work of Sakai and Sato (2007), who examined the labor market effect of caregiving requirements at the extensive margin among those aged 50 and above. Even outside Japan, only a few studies have investigated this issue. These include Dentinger and Clarkberg (2002), who examined how caregiving affects people's transition to retirement. Using US data, they found that the odds of retiring are five times higher for women caring for their husbands than noncaregiving women but find little evidence that caring for elderly parents makes caregiving children retire earlier. On the other hand, Van Houtven, Coe, and Skira (2013) found for the US that female chore caregivers are more likely to be retired while a negative effect of providing personal care is found at the extensive margin in the case of paid employment among male caregivers.

By estimating discrete-time hazard models based on panel data for Germany, Meng (2012) found that the effect of caregiving on the retirement decision is much greater than its effect on the labor market outcomes of middle-aged individuals. Having to look after a care recipient (as measured by the presence of a care recipient) is found to increase the hazard of retirement for women by 74% compared with women without caregiving responsibilities. In the case of men, it is not the mere presence of a care recipient but the time intensity of caregiving that affects their retirement decisions (Meng 2012).

Schneider *et al.* (2013) also examined the employment effects of caregiving in Austria using data on caregivers' employment plans within the next two years. Their analysis shows that the intention to exit the labor market appears to be driven by the burden of physical care provision rather than by time demands, particularly for male workers. In contrast, time-based conflicts between elderly care provision and paid work are found to be associated with a higher risk of intended job changes for female workers. However, providing care to an elderly person in need of supervision is found to make female workers less likely to exit the labor market, lending support to the argument that work can function as a respite from the burden of caregiving. In addition, Schneider *et al.* (2013) found that flexible working arrangements heighten the job and labor market attachment of female workers with caregiving responsibilities but not those of their male counterparts.

The main aim of this paper is to fill these gaps in the literature by analyzing the effect of caregiving on caregivers' retirement in the case of Japan for virtually the first time to the best of the author's knowledge. It takes a similar approach to that of Schneider et al. (2013) and examines the effect of caregiving on family caregivers' retirement intentions or more specifically on their planned retirement age. While there is a relatively large body of empirical literature on the ex post analysis of the determinants of retirement, less research has been conducted on retirement planning (Riedel, Hofer, and Wögerbauer 2015) and the present analysis makes a contribution to the literature in this respect as well. Preparing for retirement generally takes time and it would be worthwhile analyzing the implications of caregiving for caregivers' retirement plans.

Schneider *et al.* (2013) also note that employees' turnover intentions have been found to be a good proxy for actual turnover behavior in the literature (e.g., Böckerman and Ilmakunnas 2009; Steel and Ovalle 1984).

# 4. DATA AND ESTIMATION STRATEGY

#### 4.1 Data

The empirical analysis will be conducted using data from the "Preference Parameters Study" of Osaka University. This survey was conducted annually in Japan during the 2003–2013 period by the 21<sup>st</sup> Century Center of Excellence (COE) Program "Behavioral Macrodynamics Based on Surveys and Experiments" and the Global COE Project "Human Behavior and Socioeconomic Dynamics" of Osaka University. A sample of individuals aged 20–69 was drawn to be nationally representative using two-stage stratified random sampling. The sample has a panel component, although fresh observations were added in 2004, 2006, and 2009 to overcome the problem of attrition.

It would have been ideal to conduct a panel data analysis to take into account individual unobserved time-invariant heterogeneity, but unfortunately, questions regarding parental care provision were included only in the 2011 and 2013 waves and the way the key questions were phrased differed between these two waves. The advantage of using the 2011 data is that the data allow us to identify who is the main caregiver for respondents' parents and parents-in-law while data from the 2013 wave contain information only on whether or not respondents provide at least some care to their parents and/or parents-in-law. Given the importance of taking into account the intensity of caregiving when analyzing the employment effects of providing elderly care, as suggested in the literature, the empirical analysis in this paper will be conducted using data from the 2011 wave.

In addition to information on parental care provision, this survey contains basic information on respondents, including their educational attainment and employment, as well as on their households, such as household composition, consumption, income, wealth, and other socioeconomic characteristics. By exploiting this rich data set, it is possible to test the key hypotheses outlined above.

In the case of the 2011 wave, 4,934 out of 5,316 individuals completed the questionnaire (the response rate was about 93%). For the present analysis, we restrict our sample to respondents who are aged 40 or above, employed, and have at least one parent or parent-in-law alive. In other words, the sample was restricted to those who currently work and are "at risk" of retiring due to parental care responsibilities. After excluding observations with missing information on the variables used in our analysis, we were left with 970 observations. Among respondents who are already retired, there may be some who retired because of caregiving responsibilities. Given the limited information on respondents' employment or caregiving history, it was not possible to control for possible selection bias in the present analysis. For this reason, our estimates for the effect of caregiving on caregivers' retirement plans should be considered a lower bound.

# 4.2 Estimation Strategy

To investigate the effect of providing elderly care on family caregivers' retirement plans, we conduct a regression analysis of the determinants of respondents' planned retirement age. In general terms, this can be expressed as:

$$y_i = (CG_i, FC_i, X_i, \epsilon_i) \tag{1}$$

where  $y_i$  is the planned retirement age of respondent i;  $CG_i$  is a caregiving indicator variable;  $FC_i$  is a formal care usage indicator variable;  $X_i$  is a vector of demographic, socioeconomic, and employment-related variables; and  $\varepsilon_i$  is an error component.

One of the key methodological challenges of analyzing the employment effects of caregiving is that care provision is potentially endogenous to the process determining labor supply. Individuals with limited labor market opportunities or less attachment to the labor force in the first place might be more likely to become a caregiver. Similarly, individuals who would like to continue working may make use of formal care services instead of providing parental care by themselves. It is thus important to test and account for the endogeneity of the caregiving and formal care usage variables as failure to do so can lead to biased estimates of their effects on caregivers' planned retirement age. Hence, we will test for the endogeneity of the caregiving and formal care usage variables, and if they are found to be endogenous, we will adopt instrumental variable (IV) techniques. On the other hand, if they are found to be exogenous, we will resort to ordinary least squares (OLS) instead since OLS estimates are said to be more efficient than IV estimates.

While the older literature tended to ignore the potential problem of endogeneity, more recent work attempts to address it through a variety of techniques, including simultaneous equations (e.g., Johnson and Lo Sasso 2006; Wolf and Soldo 1994), difference-in-difference estimation (e.g., Spiess and Schneider 2003), dynamic panel data methods (e.g., Casado-Marín, García-Gómez, and López-Nicolás 2011; Michaud, Heitmueller, and Nazarov 2010; Moscarola 2010), and IV approaches (e.g., Bolin, Lindgren, and Lundborg 2008; Ciani 2012; Ettner 1995, 1996; Heitmueller 2007; Kotsadam 2012; Meng 2013; Nguyen and Connelly 2014). However, previous studies have reached mixed conclusions regarding the endogeneity of caregiving, with several studies finding little evidence of endogeneity and treating caregiving as exogenous (e.g., Bolin, Lindgren, and Lundborg 2008; Kotsadam 2012; Nguyen and Connelly 2014), particularly when unobserved individual fixed effects are taken into account using panel data analysis (e.g., Casado-Marín, García-Gómez, and López-Nicolás 2011; Ciani 2012; Meng 2013; Van Houtven, Coe, and Skira, 2013).

One complication we need to take into account in the present analysis is the fact that our potentially endogenous variables (i.e., the caregiving and formal care usage variables) are binary. In such a case, IV estimation using standard two-stage least squares (2SLS) estimation would generate inconsistent estimates (Wooldridge 2002). The following three-stage procedure will therefore be used instead: (i) Estimate binary response models (in this case probit models) of caregiving and formal care usage on instruments and other control variables; (ii) compute the fitted probabilities; and (iii) estimate equation (1) by IV with 2SLS using the fitted probabilities as instruments for caregiving and formal care usage (Wooldridge 2002).

Given that a greater burden of parental care tends to be shouldered by women than men and that women are generally more weakly attached to the labor market than men in Japan, as in many other countries, it would be interesting to see whether the adverse effect of caregiving on caregivers' retirement plans is greater for women than for men. Unfortunately, due to the limited number of men who serve as the main caregiver to at least one of their parents or parents-in-law in the sample, it is not possible to conduct a regression analysis using the male sample only. Instead, we will estimate equation (1) using the full and female samples and compare the regression results to infer the heterogeneous effects of caregiving on caregivers' retirement plans.

Note that our analysis is restricted to a cross-sectional data analysis due to data limitations, as discussed above. We recognize the potential limitations of this strategy and the fact that conducting a more rigorous analysis using longitudinal data to take into account individual time-invariant characteristics is left as one of the key agendas for future research.

# 4.3 Empirical Specification

The main aim of the present study is to assess the impact of elderly care provision on caregivers' retirement plans, but we will also examine the role of employment-related factors as well as individual characteristics in predicting their anticipated retirement timing.

# 4.3.1 Dependent Variable

The dependent variable is the planned retirement age of the respondent. This information was collected in the survey by asking respondents until what age they intend to work.<sup>6</sup>

### 4.3.2 Explanatory Variables

Parental care provision: The main variable of interest in the present analysis is a dummy variable capturing whether or not the respondent provides care to his/her elderly parents and/or parents-in-law. In the Preference Parameters Study, there are two questions relating to parental care provision. One of the questions asks respondents who is the main caregiver to their father, mother, father-in-law, and mother-in-law (if they are married), respectively. The other question asks respondents whether or not they and/or their spouses provide at least some care to their father, mother, father-in-law, and mother-in-law, respectively. To shed light on the importance of accounting for the intensity of caregiving, we construct two caregiving variables based on respondents' responses to these two questions. The first variable equals one if the respondent and/or his/her spouse provides some care to at least one of the respondent's parents or parents-in-law and zero otherwise. The second variable equals one only if the respondent serves as the main caregiver to at least one of his/her parents or parents-in-law and zero otherwise.

\_

It may be possible that people, for instance regular workers versus irregular workers, have a different concept of the "retirement age." However, given that the survey simply asks respondents until what age they would like to work, instead of using the word "retirement," this may be less of a concern for the current analysis. Nevertheless, we also try dropping the self-employed and irregular workers from the sample in some variants.

<sup>&</sup>lt;sup>7</sup> Unfortunately, given the way the question was phrased, we cannot separate out the respondent's role from that of his/her spouse in parental care provision in the case of married respondents.

Although some previous studies measure the intensity of caregiving in terms of a dummy variable based on an arbitrary threshold of the number of hours spent on caregiving (e.g., 20 hours per week) (e.g., Carmichael and Charles 1998; Heitmueller and Inglis 2007; Van Houtven, Coe, and Skira 2013), there is no consensus on what the level of this threshold should be. Others instead use a simpler variable such as the one used here based on self-identification as the main caregiver, which can be much more informative (e.g., Carmichael and Charles 2003; Lilly, Laporte, and Coyte 2010; Nguyen and Connelly 2014). We would expect the respondent's planned retirement age to be negatively associated with parental care provision, assuming that the substitution effect outweighs the income effect. This negative effect is expected be greater for those with higher caregiving intensity.

Formal care usage: To examine the income effect of caregiving on respondents' retirement plans, we construct a variable that equals one if formal care services (nursing homes, assisted living homes, or home helpers) play the main caregiving role for at least one of the respondent's parents or parents-in-law. If respondents need to work more to cover the cost of formal care, we would expect the coefficient on this variable to have a positive sign, providing evidence of the income effect of caregiving. However, if the financial burden of formal care usage is limited either because of the universal coverage of the LTCI program in Japan or because the financial burden is mainly borne by care recipients themselves, we may not observe a positive association between formal care usage by elderly parents and adult children's planned retirement age.

Instruments: Given that the caregiving and formal care usage variables could be endogenous, as discussed above, we will test for their endogeneity by estimating IV models. We will use parental health as an instrument for the caregiving variable. It should be directly associated with the demand for parental care and thus the caregiving status of respondents but should not directly affect respondents' retirement plans other than through the parental care path. Parental health is a commonly used instrument in the analysis of caregiving (e.g., Bolin, Lindgren, and Lundborg 2008; Crespo and Mira 2014; Ettner 1996; Kotsadam 2012; Van Houtven, Coe, and Skira, 2013). There is the possibility of the intergenerational transmission of poor health, but this can be alleviated by accounting for the health status of respondents, as commonly done in the literature (e.g., Bolin, Lindgren, and Lundborg 2008; Van Houtven, Coe, and Skira 2013). We also use parental health as an instrument for the formal care usage variable. We would expect parents/parents-in-law in poor health to be more likely to use formal care services than those in better health.

In the Preference Parameters Study, respondents were asked whether or not their parents and parents-in-law are certified as belonging to one of the seven Support/Care Levels under the LTCI system. Bigiven that this needs level is assigned based strictly on physical and mental disabilities, this variable should be a good proxy for the health status of elderly parents. Using this information, we construct a dummy variable that equals one if at least one of the respondent's parents or parents-in-law is classified as belonging to one of the five Care Levels (the degree of disability is more severe than those who are classified as belonging to one of the two Support Levels) and zero otherwise.

Under the LTCI system, the computer-aided standardized needs-assessment system categorizes people into seven levels of needs. The Care Needs Certification Board, a local committee consisting of health, medical, and welfare experts, then reviews this initial assessment and determines its appropriateness (Tsutsui and Muramatsu 2005). There are currently two levels for those who require support only (Support Levels 1 and 2) and five levels for those who require long-term care (Care Levels 1–5). This support/care level determines the amount of benefits that each person is entitled to receive.

Since the existing literature suggests that the number of siblings is a strong instrument for informal parental care provision (e.g., Bolin, Lindgren, and Lundborg 2008; Ettner 1995, 1996), we use the total number of the respondent's brothers and brothers-in-law as an additional instrument. 9 The number of brothers and brothers-in-law indicates the number of potential alternative caregivers for parents and parents-in-law. More specifically, it captures the number of daughters-in-law who traditionally take primary responsibility for elderly care in Japan, though some studies find that their role in elderly care has been declining in recent years (e.g., Hanaoka and Norton 2008). We would expect that having a greater number of brothers and brothers-in-law reduces the demand for parental care that one faces. We also tried using the total number of the respondent's sisters and sisters-in-law, but the coefficient on this variable was never significant. This suggests that the role of daughters-in-law in parental care remains relatively important in the case of Japan. Similarly, we also use as an additional instrument for parental care provision a dummy variable that equals one if the respondent and/or his/her spouse is the eldest child and zero otherwise. Given that it used to be the custom in Japan for the eldest child to take care of his/her elderly parents, we would expect being the eldest child to increase one's probability of providing parental care.

We also construct a dummy variable that equals one if both of the respondent's parents and/or parents-in-law are alive and zero otherwise as an instrument for the caregiving and formal care usage variables. If both parents and/or both parents-in-law are still alive, when one parent or parent-in-law becomes in need of care, his/her spouse is likely to serve as the main caregiver, reducing the demand for elderly care that adult children need to provide and/or the demand for formal care services. We would therefore expect having both parents and/or both parents-in-law alive to be negatively associated with the probability of providing parental care and with that of using formal care services.

Finally, we use the availability of facilities that provide institutional care in the prefecture where the respondent's parents and parents-in-law reside as an instrument for the formal care usage variable. More specifically, we use the aggregate admission capacity of facilities that provide institutional care based on data from the Survey of Institutions and Establishments for Long-term Care, which is conducted annually by the Ministry of Health, Labour and Welfare. We express this as a share of the population aged 65 and above in each prefecture. We would expect parents/parents-in-law who reside in a prefecture with a greater availability of such facilities to be more likely to use formal care services than those in a prefecture with limited availability of such facilities.

<sup>-</sup>

We define "brothers-in-law" as including only the brothers of respondents' spouses not the brothers of respondents' own siblings throughout this paper.

These institutions include facilities covered by public aid providing long-term care to the elderly, long-term care health facilities, sanatorium-type medical care facilities for the elderly requiring care, facilities that provide communal daily long-term care for dementia patients (group homes), community-based specified facilities that provide daily life long-term care, and specified facilities that provide daily life long-term care.

We use data from the 2010 Survey of Institutions and Establishments for Long-term Care given that, in our empirical analysis, we use data from the 2011 wave of the Preference Parameters Study that was conducted in January–March 2011. The data are taken from the website of the Ministry of Health, Labour and Welfare (http://www.mhlw.go.jp/toukei/saikin/hw/kaigo/service10/index.html, accessed on 22 February 2017).

Respondents' employment: Work-related factors are likely to affect people's retirement plans. We therefore include a categorical variable capturing the respondent's employment status. This variable indicates whether the respondent is (1) self-employed (including those who assist with a family business), (ii) a regular worker, or (iii) an irregular worker. 12 Given that there is in principle no retirement age for those who are self-employed, we would expect self-employed individuals to retire later than those in regular employment. As for irregular workers, given that irregular jobs tend to be characterized by more flexible working arrangements (e.g., shorter working hours), those with irregular jobs may be able to continue working until a more advanced age. It is also possible that irregular workers may not be able to retire earlier given that their jobs tend to be low paid and insecure and that they are likely to receive smaller pensions than regular workers. 13 For either reason, we would expect irregular workers to retire later than regular workers, though they may still retire earlier than self-employed individuals. We also include the respondent's wage expressed as the logarithm of his/her hourly wage. Whether higher wages induce earlier retirement is an empirical question.

**Respondents' basic characteristics:** A set of individual characteristics capturing the respondent's age, <sup>14</sup> gender, educational attainment, self-assessed health status, <sup>15</sup> and marital status as well as the number of respondents' children aged 18 or less is included. Women and those in poor health are expected to plan earlier retirement than men and those in good health, respectively. Given that unmarried (divorced, widowed, or never married) individuals do not have a spouse to count on, their planned retirement age may be older than that of married individuals.

We also include variables that reflect the respondent's preferences, such as his/her degree of time preference and risk aversion. While it is a challenge to control for these unobserved time-invariant aspects of individuals in a cross-sectional analysis, we construct variables that can serve as their proxies using the best available data from the Preference Parameters Study. <sup>16</sup> Our measure of the degree of time preference is constructed using responses to a question about whether or not respondents generally prefer getting their work done before having a good time. Our measure of the degree of risk aversion is constructed using responses to a question asking respondents to rate their behavioral pattern on a scale of 0–10 with "10" being completely in agreement with the proverb "nothing ventured, nothing gained" and "0" being completely in agreement with the proverb "a wise man never courts danger." We regard this variable as a proxy for the degree of risk preference and treat it as cardinal. We also include a variable that reflects the respondent's view toward gender roles within the household.

\_

<sup>&</sup>lt;sup>12</sup> Irregular employees include those who are working as a part-time worker, temporary worker, fixed-term worker, or dispatched worker from a temporary agency. These irregular jobs tend to be low paid and insecure in comparison with regular jobs in Japan.

Note that this employment status variable is based on information on respondents' current employment status. It is therefore possible that some irregular workers may have already retired from their regular jobs and switched to irregular jobs prior to the survey. Unfortunately, given the limited information on respondents' employment histories contained in the data, it is not possible to identify such respondents.

<sup>&</sup>lt;sup>14</sup> To examine whether the effect of caregiving on caregivers' retirement timing differs depending on the life stage of caregivers, we tried interacting the caregiving variables with age group dummies. However, the coefficients on these interaction terms were not significant in either the full or female sample.

<sup>&</sup>lt;sup>15</sup> While health is found to be an important determinant of the retirement decision, retirement is also expected to affect health (e.g., Coe and Zamarro 2011), causing a possible endogeneity problem. Using planned retirement age as our dependent variable, rather than actual retirement status or age, should help avoid such concerns (Hall and Johnson 1980).

We extract information on the degree of the respondent's time and risk preferences from data from the 2010 wave of the Preference Parameters Study as such information was not available in the 2011 wave.

This variable equals one if the respondent agrees or strongly agrees with the statement that wives should not work if their husbands earn a sufficient income. We would expect those with such a traditional view to plan to retire earlier than those without, particularly among women.

Wealth- and income-related information: Retirement timing is also inevitably closely related to the level of accumulated wealth. Unless people have sufficient financial resources for their lives after retirement, they cannot afford to retire. We therefore include a variable that indicates the level of the respondent's household wealth, which is expressed as quintiles of net worth, defined as the total amount of financial and nonfinancial assets net of liabilities. A variable that captures whether or not the respondent owns a house or an apartment is also included. Moreover, we include a variable that indicates the level of household income net of the respondent's own income, which essentially indicates the minimum amount of income that the respondent can fall back on within their household if he/she exits from the labor market. We also include a variable that indicates the share of living expenses after retirement that the respondent expects to be able to cover using public pensions (or the actual share if the respondent already receives pensions). 17 Finally, we include a dummy variable that equals one if the respondent expects to receive inheritances (including inter vivos transfers) from his/her parents and/or parents-in-law and zero otherwise. All these wealth- and income-related variables are expected to be negatively associated with the respondent's planned retirement age.

In addition to the above explanatory variables, regional dummies as well as a dummy variable for residing in a major (ordinance-designated) city are included to control for geographical variation.

# 5. EMPIRICAL RESULTS

# 5.1 Descriptive Statistics

Table 1 provides the summary statistics of the dependent and explanatory variables for the full and female samples. To obtain an overview of the characteristics of caregivers, the statistics are provided for caregivers and noncaregivers separately for each sample. Note that caregivers here refer to respondents who provide care of any intensity to at least one of their parents or parents-in-law (including those who serve as the main caregiver) while noncaregivers refer to respondents who do not provide any parental care. Unfortunately, given that this caregiving variable is constructed based on respondents' responses to the question about whether or not they and/or their spouses provide at least some parental care, we cannot determine whether it is the respondent and/or his/her spouse who provides parental care in the case of married respondents. However, since women tend to take primary responsibility for parental care in Japan, we can assume that it is primarily the respondent who provides parental care, at least in the case of the female sample. Table 1 also provides separate summary statistics for respondents who serve as the main caregiver.

<sup>&</sup>lt;sup>17</sup> We extract information on pensions from data from the 2012 wave of the Preference Parameters Study as such information was not available in the 2011 wave.

**Table 1: Summary Statistics** 

			Full S	ample		
		Care	givers			
			Main Ca	regivers	=	
		givers		nly		egivers
	Mear	n/S.D.	Mear	n/S.D.	Mear	ı/S.D.
Dependent variable						
Planned retirement age	63.94	4.91	62.84	3.95	64.22	5.38
Explanatory variables						
Caregiving variables						
Main caregiver	0.32		1.00		0.00	
Formal care usage	0.34		0.07		0.06	
Instruments						
Poor parental health	0.78		0.80		0.12	
Number of brothers and brothers-in-law	1.43	1.18	1.14	0.98	1.50	1.21
Being the eldest child	0.70		0.66		0.68	
Both parents and/or parents-in-law alive	0.41		0.25		0.63	
Availability of institutional care	3.84	0.54	3.73	0.52	3.82	0.55
Respondent's characteristics						
Age	54.51	6.12	54.32	5.58	49.86	7.03
Female	0.51		0.82		0.40	
Marital status						
Married	0.91		0.80		0.88	
Divorced/Widowed	0.03		0.09		0.06	
Never married	0.07		0.11		0.06	
No. of children aged 18 or younger	0.45	0.82	0.43	0.73	1.02	1.08
Education						
Junior high school	0.06		0.05		0.04	
High school	0.46		0.50		0.48	
Junior college	0.16		0.18		0.15	
University or above	0.32		0.27		0.34	
Poor health	0.20		0.14		0.15	
Employment	0.20		0		0.10	
Regular job	0.52		0.39		0.56	
Irregular job	0.29		0.45		0.28	
Self-employed	0.29		0.43		0.26	
Log of hourly wage	7.24	0.56	7.09	0.50	7.35	0.61
Log of flourly wage Log of other household income	0.97	0.83	1.06	0.84	0.86	0.83
•		0.03		0.04		0.63
Homeownership	0.94		0.91		0.90	
Expected receipt of inheritances	0.57	0.04	0.61	0.00	0.57	0.05
Pensions	0.54	0.21	0.57	0.20	0.47	0.25
Wealth quintiles	0.44		0.00		0.00	
1st quintile	0.14		0.09		0.22	
2nd quintile	0.19		0.25		0.19	
3rd quintile	0.20		0.18		0.22	
4th quintile	0.24		0.23		0.18	
5th quintile	0.24		0.25		0.19	
Gender roles	0.10		0.02		0.15	
Low time preference	0.53		0.57		0.50	
Risk preference	4.14	1.78	4.25	1.71	4.17	1.94
Residing in a major city	0.20		0.27		0.25	
Number of observations	13	38	4	.4	83	32

continued on next page

Table 1 continued

			Female	Sample		
		Care	givers			
			Main Ca	regivers	<del>_</del> "	
		givers n/S.D.		nly n/S.D.		regivers n/S.D.
Dependent variable						
Planned retirement age	62.48	4.21	62.50	3.69	62.26	5.53
Explanatory variables						
Caregiving variables						
Main caregiver	0.51		1.00		0.00	
Formal care usage	0.32		0.06		0.06	
Instruments						
Poor parental health	0.80		0.78		0.10	
Number of brothers and brothers-in-law	1.35	1.23	1.17	1.03	1.43	1.25
Being the eldest child	0.75		0.67		0.70	
Both parents and/or parents-in-law alive	0.32		0.17		0.63	
Availability of institutional care	3.83	0.57	3.79	0.55	3.82	0.56
Respondent's characteristics						
Age	53.63	5.90	54.44	5.83	48.31	6.12
Female	1.00		1.00		1.00	
Marital status						
Married	0.86		0.78		0.83	
Divorced/Widowed	0.06		0.11		0.11	
Never married	0.08		0.11		0.06	
No. of children aged 18 or younger	0.35	0.78	0.36	0.68	1.03	1.08
Education						
Junior high school	0.04		0.03		0.03	
High school	0.51		0.53		0.52	
Junior college	0.23		0.22		0.28	
University or above	0.23		0.22		0.17	
Poor health	0.20		0.17		0.17	
Employment						
Regular job	0.34		0.25		0.30	
Irregular job	0.48		0.56		0.53	
Self-employed	0.18		0.19		0.17	
Log of hourly wage	6.94	0.40	6.93	0.38	6.93	0.51
Log of other household income	1.35	0.79	1.16	0.82	1.36	0.76
Homeownership	0.92	00	0.92	0.02	0.89	00
Expected receipt of inheritances	0.58		0.64		0.54	
Pensions	0.54	0.21	0.59	0.20	0.45	0.26
Wealth quintiles						
1st quintile	0.14		0.11		0.22	
2nd quintile	0.25		0.25		0.20	
3rd quintile	0.15		0.17		0.20	
4th quintile	0.21		0.22		0.20	
5th quintile	0.24		0.25		0.19	
Gender roles	0.08		0.03		0.10	
Low time preference	0.55		0.56		0.53	
Risk preference	4.11	1.76	4.19	1.72	3.94	1.93
Residing in a major city	0.20	0	0.28	2	0.24	1.00
Number of observations		<u>'</u> 1				33

S.D. = standard deviations.

Source: Calculations based on data from the 2011 Preference Parameters Study.

The figures at the bottom of the table indicate that about 14% and 18% of the full and female samples are found to be engaged in parental care, respectively. In the case of the full sample, we find that about 32% of caregivers serve as the main caregiver to at least one of their parents or parents-in-law and that this figure is greater for the female sample (about 51%), underscoring the fact that women tend to make greater caregiving commitments than men.

With regard to the outcome of interest, we do not find a statistically significant difference in the planned retirement age between caregivers and noncaregivers in either the full or female sample. We find a marginally significant difference (significant at the 10% level) between main caregivers and nonmain caregivers (their statistics are not shown in the table) in the case of the full sample. It is therefore not clear whether providing care to elderly parents makes their adult children retire earlier than those without caregiving responsibilities and we will investigate this issue more rigorously through a regression analysis in the next subsection.

According to Table 1, the share of respondents whose parents and/or parents-in-law avail themselves of formal care services as their main caregiver is significantly greater among caregivers than among noncaregivers for both the full and female samples. This suggests that parental care provided by adult children and formal care services are complementary, at least if we do not take into account the intensity of care that adult children provide. Once we take into account the intensity of caregiving, we find that formal care services act as a substitute for parental care that adult children provide among respondents who serve as the main caregiver. This is shown by the fact that the share of respondents whose parents and/or parents-in-law use formal care services as their main caregiver is much lower among those who serve as the main caregiver of a parent or parent-in-law.

It is not surprising to find that the share of respondents with at least one parent or parent-in-law who is certified as requiring care under the LTCI program (i.e., having parents and/or parents-in-law in poor health) is significantly greater among caregivers than among noncaregivers for both the full and female samples. In addition, as expected, the share of respondents who have both parents and/or both parents-in-law alive is significantly lower among caregivers than among noncaregivers, and the share is particularly low among respondents who serve as the main caregiver for both samples. This suggests that spouses tend to play an important role in elderly care if they are still alive. We also find that the number of brothers and brothers-in-law that respondents have is significantly lower among those who serve as the main caregiver than among nonmain caregivers for the full sample, as expected.

Table 1 also shows that caregivers are on average older than noncaregivers, and the differences are statistically significant at the 1% level for both the full and female samples. This is as expected since the demand for caregiving tends to increase with the age of parents and hence with the age of adult children. In addition, caregivers tend to have fewer children aged 18 or younger than noncaregivers for both samples. This might be due to the fact that caregivers tend to be older and thus are less likely to have children of this age group and/or to the fact that those with relatively young children might be more preoccupied with their parenting role and thus be less likely to be engaged in elderly care than those without young children. Moreover, we find that main caregivers are less likely to have a traditional view toward gender roles, that they are likely to earn lower wages, and that they are less likely to have a regular job (more likely to have an irregular job) than nonmain caregivers in the case of the full sample. These observed differences might be caused by the fact that women with such attributes are more likely to serve as the main caregiver than men. It is interesting to find that the share of living expenses that respondents expect to be able to cover using

public pensions is significantly greater among caregivers than among noncaregivers for both samples. The differences are statistically significant at the 10% and 1% level for the full and female samples, respectively.

As for the rest of the explanatory variables listed in Table 1, we do not find a statistically significant difference between caregivers and noncaregivers for either the full or female sample.

# 5.2 Endogeneity of Caregiving and Formal Care Usage Variables

We now turn to our regression analysis of the determinants of the planned retirement age to investigate the effect of caregiving on family caregivers' retirement plans. As noted earlier, one potential problem is that our caregiving and formal care usage variables might be endogenous to the retirement decision process. We therefore test for the endogeneity of these variables by estimating IV models. To see whether the intensity of caregiving matters for the way in which caregiving affects caregivers' retirement plans, we use the (more general) caregiver as well as main caregiver variables, as explained earlier.

Following the estimation procedure outlined in Section 4.2, we first estimate binary models to obtain the fitted probabilities for providing some care to at least one parent or parent-in-law, for serving as the main caregiver to one of them, and for the use of formal care services. <sup>18</sup> We use the health status of the respondent's parents and parents-in-law, the number of the respondent's brothers and brothers-in-law, a variable that indicates whether or not the respondent and/or his/her spouse is the eldest child, a variable that indicates whether or not both parents and/or both parents-in-law of the respondent are alive, and the availability of institutional care as instruments. As shown in Table A1, a different combination of these instruments is employed for each probit model to ensure that all the instruments used in the estimation are significantly correlated with the endogenous variables.

As far as the estimated coefficients of the instruments are concerned, they all have the expected signs. The results show that having at least one parent or parent-in-law in need of care is positively associated with the probability of providing at least some parental care, of serving as the main caregiver, and of using formal care services for both the full and female samples. We also find that the number of the respondent's brothers and brothers-in-law is negatively associated with the probability of providing at least some parental care for the full sample and of serving as the main caregiver for both samples, while the fact that the respondent and/or his/her spouse is the eldest child increases their probability of providing parental care in the case of the female sample. Moreover, having both parents alive and/or both parents-in-law alive is negatively associated with the probability of providing at least some parental care, of serving as the main caregiver, and of using formal care services for both the full and female samples, as expected. Finally, the greater availability of institutional care is positively associated with the probability of using formal care services. The coefficients on these identifying instruments are jointly significant at the 1% level in all cases.

Selected regression results of the IV models as well as the specification test results are shown in the Appendix. The full results are available from the author upon request.

We then estimate the first stage of the 2SLS using the fitted probabilities obtained from the probit models as instruments. The coefficients on these predicted probabilities for providing at least some parental care, for serving as the main caregiver, and for formal care usage are estimated to be highly significant in all relevant regressions for both the full and female samples (see Table A2). In addition, the obtained F-statistic is greater than 10 in all cases, which suggests that the instruments are empirically strong according to the commonly used rule-of-thumb criterion (Staiger and Stock 1997). Despite the rejection of weak instruments, Wooldridge's (1995) score test results suggest that the null hypothesis of the exogeneity of the caregiving and formal care variables cannot be rejected in any case for both the full and female samples (see Table A3). This is consistent with the findings of some previous studies based on cross-sectional data (e.g., Bolin, Lindgren, and Lundborg 2008; Kotsadam 2012; Nguyen and Connelly 2014). We will therefore treat our caregiving and formal care usage variables as exogenous in the retirement age equation and estimate it by OLS to examine the effect of caregiving on family caregivers' retirement plans.

#### 5.3 Main Results

# 5.3.1 Full Sample

Table 2 shows the OLS regression results for the full and female samples. Model (1) examines the effect of providing at least some parental care on respondents' retirement plans while model (2) looks at the effect of serving as the main caregiver. Looking first at the regression results for the full sample, we find that providing parental care of any intensity does not have a significant effect on respondents' retirement plans. However, once we take into account the degree of respondents' caregiving intensity by using the main caregiver variable instead, we find a significant effect of caregiving on respondents' retirement. This is consistent with previous findings, which commonly find a more severe adverse employment effect of caregiving for caregivers with greater caregiving responsibilities than for those with less (e.g., Carmichael and Charles 2003; Heitmueller 2007; Kotsadam 2012; Lilly, Laporte, and Coyte 2010; Nguyen and Connelly 2014). The planned retirement age of respondents who serve as the main caregiver to at least one of their parents or parents-in-law is estimated to be, ceteris paribus, 1.2 years earlier than that of those without such responsibilities. This is a nontrivial cost to family caregivers if we consider the financial consequences of retiring earlier in terms of the loss of current income as well as the reduction in pension entitlements and possibly in lump sum severance payments.

The insignificant effect of caregiving of any intensity together with the negative and significant effect of serving as the main caregiver on retirement plans imply that, if measures such as ensuring access to formal care services allow adult children to escape from being the primary caregiver to their elderly parents and/or parents-in-law, they can help adult children to juggle their employment and caregiving responsibilities and thus save them from having to retire earlier in order to meet their caregiving responsibilities. Note that our regression results also show that the usage of formal care services by respondents' parents and/or parents-in-law has no significant effect on respondents' retirement plans. The insignificance of this effect seems to indicate that the income effect of caregiving is relatively limited in the case of Japan.

**Table 2: Ordinary Least Squares Regression Results** 

	Full Sample					
	(1	)	(2)			
Caregiver	-0.686	[0.427]				
Main caregiver			-1.162**	[0.577]		
Formal care usage	-0.126	[0.451]	-0.132	[0.424]		
Age	0.220***	[0.029]	0.221***	[0.030]		
Female	-2.674***	[0.399]	-2.612***	[0.407]		
Marital status						
(Married)						
Divorced/Widowed	1.443**	[0.601]	1.492**	[0.602]		
Never married	1.123*	[0.636]	1.172*	[0.629]		
No. of children aged 18 or younger	0.235	[0.206]	0.249	[0.205]		
Education						
(Junior high school)						
High school	0.043	[0.601]	0.065	[0.598]		
Junior college	-0.288	[0.752]	-0.291	[0.752]		
University or above	-0.111	[0.638]	-0.093	[0.636]		
Poor health	-0.419	[0.385]	-0.456	[0.384]		
Employment						
(Self-employed)						
Regular job	-4.934***	[0.528]	-4.945***	[0.527]		
Irregular job	-3.949***	[0.543]	-3.938***	[0.544]		
Log of hourly wage	0.114	[0.400]	0.132	[0.399]		
Log of other household income	-0.666***	[0.197]	-0.674***	[0.197]		
Homeownership	-0.271	[0.599]	-0.281	[0.602]		
Expected receipt of inheritances	-0.500	[0.305]	-0.492	[0.305]		
Pensions	-1.561**	[0.658]	-1.547**	[0.657]		
Wealth quintiles						
(1st quintile)						
2nd quintile	-0.480	[0.493]	-0.465	[0.493]		
3rd quintile	-1.118**	[0.466]	-1.113**	[0.467]		
4th quintile	-1.188**	[0.488]	-1.201**	[0.488]		
5th quintile	-1.135**	[0.524]	-1.158**	[0.526]		
Gender roles	-0.803*	[0.422]	-0.816*	[0.425]		
Low time preference	0.338	[0278]	0.336	[0.277]		
Risk preference	0.089	[0.081]	0.089	[0.081]		
Residing in a major city	1.262***	[0.351]	1.278***	[0.351]		
Constant	56.69***	[3.653]	56.46***	[3.655]		
No. of observations	97	'0	970			
$R^2$	0.3	51	0.3	51		

continued on next page

Table 2 continued

	Female Sample				
	(1	)	(2)		
Caregiver	-0.703	[0.615]			
Main caregiver			-1.333**	[0.649]	
Formal care usage	0.369	[0.764]	-0.007	[0.742]	
Age	0.218***	[0.053]	0.228***	[0.053]	
Female					
Marital status					
(Married)					
Divorced/Widowed	2.070**	[0.805]	2.120***	[0.807]	
Never married	1.146	[0.941]	1.206	[0.922]	
No. of children aged 18 or younger	0.196	[0.320]	0.220	[0.321]	
Education					
(Junior high school)					
High school	-1.143*	[0.852]	-1.318	[0.843]	
Junior college	-1.767*	[0.960]	-1.671*	[0.951]	
University or above	-0.905	[1.043]	-0.824	[1.040]	
Poor health	-0.154	[0.660]	-0.152	[0.661]	
Employment					
(Self-employed)					
Regular job	-4.767***	[0.876]	-4.807***	[0.868]	
Irregular job	-4.155***	[0.806]	-4.142***	[0.803]	
Log of hourly wage	0.892	[0.682]	0.877	[0.682]	
Log of other household income	-0.608*	[0.315]	-0.650**	[0.316]	
Homeownership	0.295	[0.903]	0.307	[0.907]	
Expected receipt of inheritances	-0.582	[0.488]	-0.544	[0.489]	
Pensions	-0.892	[1.105]	-0.808	[1.096]	
Wealth quintiles					
(1st quintile)					
2nd quintile	-1.108	[0.820]	<b>–</b> 1.115	[0.819]	
3rd quintile	-0.814	[0.877]	-0.842	[0.878]	
4th quintile	-1.630*	[0.868]	-1.656*	[0.866]	
5th quintile	-1.844**	[0.890]	-1.903**	[0.888]	
Gender roles	-1.820*	[0.987]	-1.849*	[0.990]	
Low time preference	0.548	[0.468]	0.534	[0.464]	
Risk preference	0.224	[0.137]	0.229*	[0.137]	
Residing in a major city	0.822	[0.574]	0.868	[0.574]	
Constant	48.35***	[6.068]	47.86***	[6.060]	
No. of observations	40	)4	40	)4	
$R^2$	0.2	78	0.2	81	

Note: \*\*\*, \*\*, and \* denote statistical significance at the 1%, 5%, and 10% levels. Robust standard errors are in parentheses. Regional dummies are included in all regressions.

Source: Estimation based on data from the 2011 Preference Parameters Study.

		(1.0.0)	J J					
	Full Sample				Female Sample			
	(	1)	(2	2)	(	1)	(2	2)
Caregiver/Main caregiver	-0.686 [0.427]	-0.741 [0.515]	-1.162** [0.577]	-1.461** [0.582]	-0.703 [0.615]	-0.871 [0.678]	-1.333** [0.649]	-1.565** [0.661]
*Formal care usage		0.220 [0.896]		4.056** [1.642]		0.734 [1.432]		3.444* [1.917]
Formal care usage	0.126 [0.451]	0.042 [0.589]	-0.132 [0.424]	-0.265 [0.427]	0.369 [0.764]	0.035 [1.107]	-0.007 [0.742]	-0.178 [0.756]
No. of observations	970	970	970	970	404	404	404	404
$R^2$	0.351	0.351	0.351	0.353	0.278	0.279	0.281	0.282

Table 3: Ordinary Least Squares Regression Results (Role of Formal Care Services)

Note: \*\* and \* denote statistical significance at the 5% and 10% levels. Robust standard errors are in parentheses. The set of explanatory variables listed in Table 1 as well as regional dummies are also included in the regressions.

Source: Estimation based on data from the 2011 Preference Parameters Study.

To further examine the possibility of formal care services attenuating or eliminating the adverse effect of caregiving on caregivers' planned retirement age, we try interacting the caregiving variables with the formal care usage variable. If the coefficient on the interaction term is positive and significant, this will constitute additional evidence that formal care services help alleviate or eliminate the negative effect of caregiving on caregivers' retirement plans. It should, however, be noted that our formal care usage variable equals one if formal care services play the main caregiving role for at least one of the respondent's parents or parents-in-law. Hence, in the case of model (2) where the main caregiver variable is used, the interaction term indicates a situation where the respondent takes primary responsibility for providing care to at least one of his/her parents or parents-in-law while formal care services play the main caregiving role for at least one of the rest of his/her parents or parents-in-law (i.e., there is more than one person in need of care in the household). Table 3 shows the relevant results for the full and female samples.<sup>19</sup>

As far as the results for the full sample are concerned, the coefficient on the interaction term between the caregiver variable and the formal care usage variable is found not to be statistically significant. However, the coefficient on the interaction term between the main caregiver variable and the formal care usage variable is positive and significant, and in fact the magnitude of this coefficient is larger than that of the coefficient on the main caregiver variable. The estimates indicate that serving as the main caregiver to at least one of his/her parents or parents-in-law reduces, *ceteris paribus*, the planned retirement age of the respondent by 1.5 years. However, if formal care services take primary responsibility for parental care for at least one of the rest of his/her parents or parents-in-law, serving as the main caregiver increases his/her planned retirement age by 2.6 years instead. These results provide support for the view that formal care services attenuate the need for adult children to retire earlier due to caregiving commitments.

However, given that the positive effect of formal care usage more than offsets the negative effect of serving as the main caregiver on caregivers' retirement plans, the use of formal care services may not only attenuate the adverse effect on retirement of taking primary responsibility for parental care but also cause additional effects, namely the income effect of caregiving. It may be costly to have more than one parent or

20

The rest of the regression results are very close to those reported in Table 2 in terms of the sign, significance, and size of the coefficients. The full regression results are available from the author upon request.

parent-in-law in need of care in the household, and this may require adult children to delay their retirement in order to meet the financial cost of formal care services.

As for the rest of the regression results for the full sample, as expected, it is found that women expect to retire earlier than men and that unmarried (divorced, widowed, or never married) respondents expect to retire later than their married counterparts. The latter might be due to the fact that unmarried people may face greater financial insecurity than married people, and hence they may not be able to afford to retire earlier. We also find that respondents with a regular or irregular job expect to retire earlier than those who are self-employed, as expected. The size of the negative effect on the planned retirement age is greater for regular employment than for irregular employment, implying that the extent to which regular workers retire earlier than the self-employed is greater than in the case of irregular workers, *ceteris paribus*. As in the case of unmarried people, irregular workers may not be able to afford to retire earlier given that they may face greater financial insecurity in their later years than regular workers. On the other hand, irregular jobs may have more flexibility in terms of working conditions and it might therefore be easier for irregular workers to continue working until a more advanced age than regular workers.

As expected, the level of household income net of respondents' personal incomes as well as the level of household wealth are negatively associated with respondents' planned retirement age. In addition, respondents who expect to receive inheritances (including *inter vivos* transfers) are found to retire earlier than those without such expectations. We also find that respondents who expect to receive a greater amount of public pensions relative to their living expenses are likely to retire earlier. These findings suggest that respondents with greater financial resources can afford to retire earlier than those with less, as expected. It is interesting to find that those with traditional views toward gender roles plan to retire earlier than those without.

#### 5.3.2 Female Sample

We now turn to the results for the female sample (see Table 2). Even when we restrict the sample to female respondents only, we do not find a significant effect of providing parental care of any intensity on respondents' retirement plans. However, we find a slightly larger negative effect of serving as the main caregiver on the planned retirement age for the female sample than for the full sample. In the case of female respondents, those who are the main caregiver to at least one of their parents or parents-in-law are, *ceteris paribus*, estimated to plan to retire 1.3 years earlier than those who are not. The fact that caregiving is found to have a greater adverse effect for the female sample than for the full sample underscores the fact that women are likely to bear a greater share of the burden of parental care than men, which is consistent with previous findings (e.g., Ettner 1996; Yamada and Shimizutani 2015). On the other hand, as in the case of the full sample, we do not find a significant effect of formal care usage on respondents' retirement plans.

As for the results for the interaction terms between the caregiving variables and the formal care usage variable (see Table 3), the results are similar to those for the full sample. If we use the general caregiver variable, the coefficient on the interaction term is again not statistically significant. Table 3 also shows that if female respondents serve as the main caregiver to at least one of their parents or parents-in-law, their planned retirement age decreases by 1.6 years. In contrast, if formal care services play the main caregiving role for at least one of the rest of their parents or parents-in-law, serving as the main caregiver to their parents and/or parents-in-law increases female caregivers' planned retirement age by 1.9 years. These results are similar to those obtained for the full sample, but the adverse effect of caregiving on caregivers'

retirement plans is larger, and the attenuation effect of formal care usage on the negative effect of caregiving (and/or the income effect of caregiving) is smaller, for the female sample than for the full sample.

The rest of the results are similar to those obtained for the full sample except that the amount of public pensions that respondents expect to receive relative to their living expenses no longer has a significant effect on the planned retirement age in the case of the female sample.

Table 4: Ordinary Least Squares Regression Results (for Wage Earners only)

	Full S	ample	Female Sample			
	Regular and Irregular Workers Only	Regular Workers Only	Regular and Irregular Workers Only	Regular Workers Only		
Main caregivers	-0.904 [0.605]	-0.765 [0.922]	-1.351* [0.720]	-0.217 [1.179]		
Irregular job	1.187*** [0.432]		0.966 [0.680]			
No. of observations	808	538	333	124		
$R^2$	0.260	0.201	0.243	0.328		

Note: \*\*\* and \* denote statistical significance at the 1% and 10% levels. Robust standard errors are in parentheses. The set of explanatory variables listed in Table 1 as well as regional dummies are also included in the regressions.

Source: Estimation based on data from the 2011 Preference Parameters Study.

# 5.3.3 Caregiving Effect and Employment Status

In the case of Japan, the adverse economic consequences of retiring earlier due to caregiving commitments is likely to be greater for regular workers than for irregular workers due to the limited job re-employment opportunities for regular employment in this country. The concept of retirement may also differ between regular (or irregular) workers and those who are self-employed. We would therefore like to examine whether the adverse effect of serving as the main caregiver remains significant even if we remove the self-employed or irregular workers from the estimation sample. To investigate this, we restrict our full and female samples to (i) wage earners only (regular and irregular workers only, excluding the self-employed) and (ii) regular workers only. The relevant results are reported in Table 4.<sup>20</sup>

Once we restrict our sample to wage earners or regular workers only, the effect of serving as the main caregiver on caregivers' retirement plans becomes insignificant in the case of the full sample. As for the female sample, the negative effect of serving as the main caregiver on caregivers' retirement plans is still observed even if we restrict our sample to wage earners only, but it becomes insignificant if we further restrict our sample to regular workers only. These results imply that the adverse effect of caregiving on caregivers' retirement plans is mainly observed for the self-employed in the case of men and for the self-employed and irregular workers in the case of women. Since working as a self-employed worker tends to involve greater responsibilities and longer working hours than regular employment, it might be more difficult for adult

22

The rest of the regression results are very close to those reported in Table 2 in terms of the sign, significance, and size of the coefficients. The full regression results are available from the author upon request.

children to combine their work with primary caregiving responsibilities if they are self-employed.

The above results also suggest that the adverse effect of caregiving on caregivers' retirement timing is relatively limited in the case of regular workers for both men and women in the case of Japan. This might be due to the fact that regular workers try not to retire earlier, if possible, for caregiving responsibilities given that the opportunity cost of retiring earlier is relatively high for them in terms of the consequent reduction in pension entitlements as well as lump sum severance payments in comparison with irregular workers or those who are self-employed.

Nevertheless, since more than half of the female sample is engaged in irregular employment, the negative and significant effect of caregiving on intensive caregivers' retirement plans for regular and irregular workers among women remains a cause for concern for the Japanese government, which has recently been trying to promote the labor supply of women as well as the elderly to address the shrinkage of the working-age population. Since no indications of endogeneity were detected, the provision of elderly care is unlikely to reflect the fact that those who provide elderly care face limited labor market opportunities or are less attached to the labor market. In addition, the fact that the adverse effect of caregiving on adult children's retirement was found to be greater for women than for men suggests that caregiving may put disproportionately more women at risk of having to retire earlier in order to meet their caregiving responsibilities, which could also put them at risk of living in poverty during their later years.

On the other hand, the insignificant effect of caregiving of any intensity on caregivers' planned retirement age seems to indicate that adult children may still be able to juggle their employment and caregiving responsibilities and thus may not have to retire earlier to meet their caregiving responsibilities as long as they do not have to take primary responsibility for parental care, i.e., if they perform only a supplementary role in parental care. This finding underscores the important role that the availability of formal care services can play not only in determining the amount of parental care that adult children need to provide but also in the way caregiving responsibilities affect adult children's labor market outcomes (e.g., Kotsadam 2012; Michaud, Heitmueller, and Nazarov 2010).

Our regression results indeed suggest that the use of formal care services may help attenuate or eliminate the adverse effect of taking primary responsibility for parental care on adult children's retirement timing. This is in line with the argument that more formal care leads to more choices for family members and therefore less adverse effects on their employment (Kotsadam 2012). However, given that the positive effect of formal care usage is found to more than offset the negative effect of serving as the main caregiver on caregivers' retirement plans, it seems more plausible to assume that the use of formal care services also generates an income effect of caregiving on caregivers' planned retirement age.

Our regression results indicate that the income effect of caregiving on adult children's retirement is relatively limited in most cases. This may be partly due to the existence of the LTCI program in Japan, which allows those certified as requiring care to avail themselves of formal care services at relatively low cost, and/or to the fact that the cost of formal care services may be borne by elderly parents themselves and not by their children. Nevertheless, the finding of the possible presence of an income effect in cases in which respondents act as the main caregiver to at least one of their parents or parents-in-law while formal care services take primary responsibility for parental care for the rest of their parents or parents-in-law suggests that the financial burden of

formal care services may require caregivers to postpone their retirement when there is more than one person in need of care in the household.

# 6. CONCLUSIONS

In this paper, we used microdata from a Japanese survey to examine the effect of providing parental care on adult children's retirement plans, a topic that has received relatively little attention so far. Our regression results show no significant effect of caregiving on family caregivers' planned retirement age if we do not take into account caregiving intensity. However, once we account for the intensity of caregiving, we find that serving as the main caregiver to parents and/or parents-in-law has a negative and significant effect on adult children's planned retirement age. The comparison of the estimates based on the full sample with those based on the female sample also indicates that the adverse effect of providing parental care on adult children's retirement plans tends to be greater for female caregivers than for their male counterparts. On the other hand, the negative effect of caregiving on caregivers' retirement timing is found to be relatively limited for regular and irregular workers in the case of men and for regular workers in the case of women.

These findings seem to suggest that, if ensuring access to formal care services can allow family members to escape from taking primary responsibility for providing care to their parents and/or parents-in-law, it can help adult children to reconcile their paid work with caregiving responsibilities, thereby alleviating the adverse effect of caregiving on their retirement plans. Our regression results indeed indicate that formal care services could help alleviate or eliminate the negative effect of serving as the main caregiver on adult children's retirement plans. However, our results also suggest the possibility that the use of formal care services could also generate an income effect of caregiving. While our findings suggest a limited income effect of caregiving in most cases, we find some evidence of the possibility that the financial burden of formal care services may require caregivers to postpone requirement in some cases, for instance when there is more than one person in need of care in the household.

Our analysis is certainly not without its caveats. Data limitations did not allow us to control fully for individual unobserved time-invariant characteristics. To confirm our findings, it is necessary to conduct a more rigorous analysis using longitudinal data to take into account such effects. Moreover, we have examined the employment effects of caregiving only at the extensive margin, but analyzing it at the intensive margin is equally important as people may reduce their workload before exiting from the labor market as part of their transition into retirement. Effects at the intensive margin may have implications for current earnings as well as retirement income that could affect the livelihood of family caregivers even long after caregiving responsibilities end.

Despite these limitations, this paper is one of the first to examine the effect of providing parental care on adult children's retirement plans in Japan and generated some key findings that have important policy implications. Our finding that intensive caregivers are likely to plan to retire earlier than those without such responsibilities suggests that caregiving responsibilities are likely to impose long-term financial costs on family caregivers in terms of the loss in current earnings as well as reduced pension entitlements. This in turn implies that relying on family members to provide elderly care has significant potential costs to society in terms of not only reduced labor supply but also a reduction in income tax revenue as well as contributions to pension funds. It is therefore critical to ensure that family members have ways to reconcile their paid work with caregiving requirements, for instance through the availability of formal care

services as well as creating flexible working environments. Unless the cost of informal elderly care for family members in terms of, among other things, earlier retirement is recognized and addressed, the increasing burden of elderly care will pose a serious challenge to the ongoing efforts of the Japanese government to promote the labor market participation of women and the elderly as a solution to the shrinkage of the working-age population.

# REFERENCES

- Bauer, J.M., and A. Spousa-Poza. 2015. "Impacts of Informal Caregiving on Caregiver Employment, Health, and Family," *Journal of Population Ageing* 8(3), 113–145.
- Berecki-Gisolf, J., J. Lucke, R. Hockey, and A. Dobson. 2008. "Transitions into Informal Caregiving and Out of Paid Employment of Women in Their 50s," *Social Science & Medicine* 67(1), 122–127.
- Bittman, M., T. Hill, and C. Thomson. 2007. "The Impact of Caring on Informal Carers' Employment, Income and Earnings: A Longitudinal Approach," *Australian Journal of Social Issues* 42(2), 255–272.
- Böckerman, P., and P. Ilmakunnas. 2009. "Job Disamenities, Job Satisfaction, and Quit Intentions, and Actual Separations: Putting the Pieces Together," *Industrial Relations: A Journal of Economy and Society* 48(1), 73–96.
- Bolin, K., B. Lindgren, and P. Lundborg. 2008. "Your Next of Kin or Your Own Career? Caring and Working among the 50+ of Europe," *Journal of Health Economics* 27(3), 718–738.
- Carmichael, F., and S. Charles. 1998. "The Labour Market Costs of Community Care," Journal of Health Economics 17(6), 747–765.
- ——. 2003. "The Opportunity Costs of Informal Care: Does Gender Matter?" *Journal of Health Economics* 22(5), 781–803.
- Casado-Marín, D., P. García-Gómez, and A. López-Nicolás. 2011. "Informal Care and Labour Force Participation among Middle-aged Women in Spain," *SERIEs* 2(1), 1–29.
- Ciani, E. 2012. "Informal Adult Care and Caregivers' Employment in Europe," *Labour Economics* 19(2), 155–164.
- Coe, N.B., and G. Zamarro. 2011. "Retirement Effects on Health in Europe," *Journal of Health Economics* 30(1), 77–86.
- Crespo, L., and P. Mira. 2014. "Caregiving to Elderly Parents and Employment Status of European Mature Women," *Review of Economics and Statistics* 96(4), 693–709.
- Dentinger, E., and M. Clarkberg. 2002. "Informal Caregiving and Retirement Timing among Men and Women: Gender and Caregiving Relationships in Late Midlife," *Journal of Family Issues* 23(7), 857–879.
- Do, Y.K. 2008. "The Effect of Informal Caregiving on Labor Market Outcomes in South Korea," *Asia Health Policy Program Working Paper*, No. 1, Walter H. Shorenstein Asia-Pacific Research Center, Stanford University.
- Ettner, S.L. 1995. "The Impact of 'Parent Care' on Female Labor Supply Decisions," *Demography* 32(1), 63–80.
- ——. 1996. "The Opportunity Costs of Elder Care," *Journal of Human Resources* 31(1), 189–205.
- Hall, A., and T.R. Johnson. 1980. "The Determinants of Planned Retirement Age," *Industrial and Labor Relations Review* 33(2), 241–254.
- Hanaoka, C., and E.C. Norton. 2008., "Informal and Formal Care for Elderly Persons: How Adult Children's Characteristics Affect the Use of Formal Care in Japan," *Social Science & Medicine* 67(6), 1002–1008.

- Heitmueller, A. 2007. "The Chicken or the Egg? Endogeneity in Labour Market Participation of Informal Carers in England," *Journal of Health Economics* 26(3), 536–559.
- Heitmueller, A., and K. Inglis. 2007. "The Earnings of Informal Carers: Wage Differentials and Opportunity Costs," *Journal of Health Economics* 26(4), 821–841.
- Johnson, R.W., and A.T. Lo Sasso. 2006. "The Impact of Elder Care on Women's Labor Supply," *Inquiry* 43(3), 195–210.
- Kan, M., and S. Kajitani. 2014., "The Impact of Public Long-term Care Insurance on Time Spent on Informal Care among At-home Caregivers: Findings from Japanese Micro Data," (in Japanese) *The Economic Review* 65(4), 345–361.
- Kotsadam, A. 2012. "The Employment Costs of Caregiving in Norway," *International Journal of Health Care Finance and Economics* 12(4), 269–283.
- Leigh, A. 2010. "Informal Care and Labor Market Participation," *Labour Economics* 17(1), 140–149.
- Lilly, M.B., A. Laporte, and P.C. Coyte. 2007. "Labor Market Work and Home Care's Unpaid Caregivers: A Systematic Review of Labor Force Participation Rates, Predictors of Labor Market Withdrawal, and Hours of Work," *Milbank Quarterly* 85(4), 641–690.
- ———. 2010. "Do They Care too Much to Work? The Influence of Caregiving Intensity on the Labour Force Participation of Unpaid Caregivers in Canada," *Journal of Health Economics* 29(6), 895–903.
- Long, S.O., R. Campbell, and C. Nishimura. 2009. "Does It Matter Who Cares? A Comparison of Daughters versus Daughters-in-Law in Japanese Elder Care," *Social Science Japan Journal* 12(1), 1–21.
- Meng, A. 2012. "Informal Caregiving and the Retirement Decision," *German Economic Review* 13(3), 307–330.
- ———. 2013. "Informal Home Care and Labor-force Participation of Household Members," *Empirical Economics* 44(2), 959–979.
- Michaud, P., A. Heitmueller, and Z. Nazarov. 2010. "A Dynamic Analysis of Informal Care and Employment in England," *Labour Economics* 17(3), 455–465.
- Moscarola, F.C. 2010. "Informal Caregiving and Women's Work Choices: Lessons from the Netherlands," *Labour* 24(1), 93–105.
- National Institute of Population and Social Security Research 2012. *Population Projections for Japan (January 2012): 2011 to 2060.* http://www.ipss.go.jp/site-ad/index\_english/esuikei/gh2401e.asp (accessed 18 September 2015).
- Nguyen, H.T. and L.B. Connelly. 2014. "The Effect of Unpaid Caregiving Intensity on Labour Force Participation: Results from a Multinomial Endogenous Treatment Model," Social Science & Medicine 100, 115–122.
- Niimi, Y. 2016. "The 'Costs' of Informal Care: An Analysis of the Impact of Elderly Care on Caregivers' Subjective Well-being in Japan," *Review of Economics of the Households* 14(4), 779–810.
- Oshio, T. and E. Usui. 2016. "Informal Parental Care and Female Labour Supply in Japan," *Applied Economics Letters* doi: 10.1080/13504851.2016.1217303.

- Riedel, M., H. Hofer, and B. Wögerbauer. 2015. "Determinants for the Transition from Work into Retirement in Europe," *IZA Journal of European Labor Studies* 4(4), doi: 10.1186/s40174-014-0027-5.
- Sakai, T. and K. Sato. 2007. "Kaigo ga Koreisha no Shugyo/Taishoku Kettei ni Oyobosu Eikyo (in Japanese) (The Impact of Caregiving on the Employment and Retirement Decisions of the Aged)," *JCER Economic Journal* 56, Japan Center for Economic Research.
- Schneider, U., B. Trukeschitz, R. Mühlmann, and I. Ponocny. 2013. "Do I Stay or Do I Go?' Job Change and Labor Market Exit Intentions of Employees Providing Informal Care to Older Adults," *Health Economics* 22(10), 1230–1249.
- Shimizutani, S., W. Suzuki, and H. Noguchi. 2008. "The Socialization of At-home Elderly Care and Female Labor Market Participation: Micro-level Evidence from Japan," *Japan and the World Economy* 20(1), 82–96.
- Spiess, C.K., and A.U. Schneider. 2003. "Interactions between Care-giving and Paid Work Hours among European Midlife Women, 1994 to 1996," *Ageing and Society* 23(1), 41–68.
- Staiger, D., and J.H. Stock. 1997. "Instrumental Variables Regression with Weak Instruments," *Econometrica* 65(3), 557–586.
- Steel, R.P., and N.K. Ovalle. 1984. "A Review and Meta-analysis of Research on the Relationship between Behavioral Intentions and Employee Turnover," *Journal of Applied Psychology* 69(4), 673–686.
- Sugawara, S., and J. Nakamura. 2014. "Can Formal Elderly Care Stimulate Female Labor Supply? The Japanese Experience," *Journal of the Japanese and International Economies* 34, 98–115.
- Tsutsui, T., and N. Muramatsu. 2005. "Care-Needs Certification in the Long-Term Care Insurance System of Japan," *Journal of the American Geriatrics Society* 53(3), 522–527.
- Van Houtven, C.H., N.B. Coe, and M.M. Skira. 2013. "The Effect of Informal Care on Work and Wages," *Journal of Health Economics* 32(1), 240–252.
- Wolf, D.A., and B.J. Soldo. 1994. "Married Women's Allocation of Time to Employment and Care of Elderly Parents," *Journal of Human Resources* 29(4), 1259–1276.
- Wooldridge, J.M. 1995. "Score Diagnostics for Linear Models Estimated by Two Stage Least Squares." In: *Advances in Econometrics and Quantitative Economics: Essays in Honor of Professor C. R. Rao.* edited by Maddala, G.S., T.N. Srinivasan and P.C.B. Philips Oxford: Blackwell.
- ——. 2002. Econometric Analysis of Cross Section and Panel Data. Cambridge: The MIT Press.
- Yamada, H., and S. Shimizutani. 2015. "Labor Market Outcomes of Informal Care Provision in Japan," *Journal of the Economics of Ageing* 6, 79–88.

# **APPENDIX: IV ESTIMATION RESULTS**

Table A1: First-stage Binary (Probit) Model Estimation

			, ,				
			Full Sa	ample			
	Careg	jivers	Main Ca	regivers	Formal Care		
Poor parental health	0.422***	[0.038]	0.132***	[0.025]	0.346***	[0.037]	
Number of brothers and brothers-in-law	-0.019**	[800.0]	-0.015**	[0.006]			
Being the eldest child							
Both parents and/or parents-in-law alive	-0.039*	[0.022]	-0.043***	[0.014]	-0.036*	[0.020]	
Availability of institutional care							
Jointly significant at 1%	•	1	<b>✓</b>	•	•	/	
No. of observations	97	70	97	<b>'</b> 0	97	70	
LR chi2	302	57	160	.55	246.03		
Prob > chi2	0.0	00	0.0	00	0.000		
Log likelihood	-24	5.50	-98.81		-201.01		
Pseudo R <sup>2</sup>	0.3	81	0.4	48	0.380		
			Female	Sample			
	Caregivers		Main Ca	regivers	Formal Care		
Poor parental health	0.529***	[0.057]	0.225***	[0.043]	0.338***	[0.048]	
Number of brothers and brothers-in-law			-0.020*	[0.012]			
Being the eldest child	0.060**	[0.028]					
Both parents and/or parents-in-law alive	-0.101***	[0.036]	-0.128***	[0.032]	-0.071**	[0.030]	
Availability of institutional care					0.053**	[0.027]	
Jointly significant at 1%	•	1	✓		✓		
No. of observations	40	)4	404		404		
LR chi2	195	.34	122.38		139	.92	
Prob > chi2	0.0	00	0.000		0.000		
Log likelihood	-90	.14	-60	.20	-67.00		
Pseudo R <sup>2</sup>	0.5	20	0.504		0.511		

Note: The estimated results are reported in terms of average marginal effects. \*\*\*, \*\*\*, and \* denote statistical significance at the 1%, 5%, and 10% levels. Standard errors are in parentheses. The set of explanatory variables listed in Table 1 as well as regional dummies are also included in the regressions.

Source: Estimation based on data from the 2011 Preference Parameters Study.

Table A2: First-stage Results for IV Estimation (2SLS)

	Full Sample					
	IV	<sup>'</sup> (1)	IV	<sup>'</sup> (2)		
	Caregiver	Formal Care Use	Main Caregiver	Formal Care Use		
Predicted probability of being caregivers	1.104*** [0.185]	-0.024 [0.183]				
Predicted probability of being main caregivers			0.952*** [0.151]	0.004 [0.165]		
Predicted probability of formal care usage	-0.117 [0.223]	1.040*** [0.230]	-0.013 [0.055]	1.011*** [0.108]		
No. of observations	970	970	970	970		
Adjusted R <sup>2</sup>	0.359	0.304	0.261	0.304		
Instrumental variable strength (F-statistics)	102.07***	64.92***	24.22***	64.51***		

		Female	Sample	
	IV	<sup>'</sup> (1)	IV	<sup>'</sup> (2)
	Caregiver	Formal Care Use	Main Caregiver	Formal Care Use
Predicted probability of being caregivers	1.040*** [0.136]	0.059 [0.133]		
Predicted probability of being main caregivers			1.021*** [0.141]	–1.182 [0.135]
Predicted probability of formal care usage	-0.042 [0.195]	0.889*** [0.180]	-0.102 [0.099]	1.031*** [0.117]
No. of observations	404	404	404	404
Adjusted R <sup>2</sup>	0.502	0.376	0.357	0.382
Instrumental variable strength (F-statistics)	84.49***	35.98***	28.78***	41.82***

Note: \*\*\* denotes statistical significance at the 1% level. Robust standard errors are in parentheses. The set of explanatory variables listed in Table 1 as well as regional dummies are also included in the regressions.

Source: Estimation based on data from the 2011 Preference Parameters Study.

Table A3: Second-stage Results for IV Estimation (2SLS)

	Full Sample				Female Sample			
	IV	(1)	IV	(2)	IV	(1)	IV	(2)
Caregiver	-1.003	[1.537]			-0.255	[1.296]		
Main caregiver			-0.947	[1.428]			-2.207*	[1.245]
Formal care services	0.802	[1.850]	0.011	[0.914]	-0.523	[1.875]	0.017	[1.252]
No. of observations	9	70	970		404		404	
Wald chi2	472	2.24	474.34		165.34		166.09	
Prob > chi2	0.0	000	0.000		0.000		0.000	
$R^2$	0.350		0.351		0.276		0.2	279
Exogeneity test (chi2 (p-values))**	0.312	(0.856)	0.151	(0.927)	0.329 (0.848)		0.928 (	(0.629)

Note: \* denotes statistical significance at the 10% level. Robust standard errors are in parentheses. The set of explanatory variables listed in Table 1 as well as regional dummies are also included in the regressions.

Source: Estimation based on data from the 2011 Preference Parameters Study.

<sup>\*</sup> Wooldridge's (1995) score test